Author’s response to reviews

Title: Use of a public park for physical activity in the Caribbean. Evidence from a mixed methods study in Jamaica

Authors:

Colette Cunningham-Myrie (colette.cunninghammyrie@uwimona.edu.jm)
Tamika Royal-Thomas (tamika.royalthomas@gmail.com)
Althea Bailey (althea.bailey02@uwimona.edu.jm)
Jeanette Gustat (gustat@tulane.edu)
Katherine Theall (ktheall@tulane.edu)
Joy Harrison (joy.harrison02@uwimona.edu.jm)
Marvin Reid (marvin.reid@uwimona.edu.jm)

Version: 4 Date: 24 Jun 2019

Author’s response to reviews:

Response to reviewer’s comments from previous manuscript submission

ID number (PUBH-D-18-02510R3)

We deeply appreciate the editor’s comments on our revised submission. We have incorporated and addressed all concerns in this latest revised manuscript, and as described below.

Assistant Editor Comments:

Image attribution Figure 1
Thank you for providing image attribution for Figure 1 (Google Earth (2019)). As maps from Google Earth are currently under license, please provide written evidence of permissions obtained to publish the map from Google Earth.

If permissions to publish the image have not been obtained, the image needs to be removed.

Response: We were unaware that maps from Google Earth were under license. We have decided to remove the map and made edits to the manuscript by reassigning new numbering for the remaining figures on page 5 lines 12-15 and page 12 line 6. Edits were also made to the legends of the figures on pages 32 to 35, to reflect the change in the numbering of these figures. We hope these changes are now satisfactory.